Author's response to reviews

Title: Patterns and correlates of physical activity: a cross-sectional study in urban Chinese women

Authors:

Adriana L Jurj (adriana.jurj@sc.edu)
Wanqing Wen (wangqing.wen@vanderbilt.edu)
Yu-Tang Gao (ytgao@vip.sina.com)
Charles E Matthews (charles.matthews@vanderbilt.edu)
Gong Yong (gong.yang@vanderbilt.edu)
Hong-Lan Li (hlhan@online.sh.cn)
Wei Zheng (wei.zheng@vanderbilt.edu)
Xiao-Ou Shu (xiao-ou.shu@vanderbilt.edu)

Version: 3 Date: 13 August 2007

Author's response to reviews: see over
August 13, 2007

Editorial Team
BMC Public Health

Re: IMD07-0591

Dear Editorial Team:

Thank you for your email of August 8, 2007 requesting formatting changed to our manuscript. We have carefully addressed all the points raised in the email to ensure that the text and tables are properly formatted. We hope our revised manuscript meets with your approval for publication.

Sincerely,

Xiao Ou Shu, M.D., Ph.D.
Vanderbilt University Medical Center
Vanderbilt Epidemiology Center
Institute for Medicine and Public Health
2525 West End Avenue, Suite 600
Nashville, TN 37203-1738
Tel: 615-936-0713
Fax: 615-936-8291
E-mail: xiao-ou.shu@vanderbilt.edu